Author’s response to reviews

Title: Antinociceptive tolerance to NSAIDs in the anterior cingulate cortex is mediated via endogenous opioid mechanism

Authors:

Merab Tsagareli (m.tsagareli@biomedicine.org.ge)
Nana Tsiklauri (nanatsiklauri1812@gmail.com)
Natia Pirkulashvili (nat10a@yahoo.com)
Ivliane Nozadze (nozadzelia@yahoo.com)
Marina Nebieridze (mnebo@mail.ru)
Gulnazi Gurtskaia (gurtskaiaguliko@yahoo.com)
Elene Abzianidze (eleneabzianidze@gmail.com)

Version: 2 Date: 09 Nov 2017

Author’s response to reviews:

I would like to express my sincere appreciation to both Reviewers, Prof. Jun-Xu Li (Reviewer 1), and Prof. Ahmet Ulugol (Reviewer 2) for their positive comments, notes and decisions.

I have put an Abbreviations section in the end of the manuscript.

A native speaker of English polished English grammar.

I wish to express my special thanks to Chief-Editor, Prof. Benjamin Ragen for his very kind attitude and support.